Author’s response to reviews

Title: Use, applicability and reliability of Depth of Hypnosis monitors in children - A survey among members of the European Society for Paediatric Anaesthesiology

Authors:
Yuen Cheung (y.m.cheung@erasmusmc.nl)
Gail Scoones (g.scoones@erasmusmc.nl)
Robert Jan Stolker (r.stolker@erasmusmc.nl)
Frank Weber (f.weber@erasmusmc.nl)

Version: 3 Date: 12 Mar 2018

Author’s response to reviews:

Dear dr. Faraoni,

I would like to thank you once again for your help to improve our manuscript. In the revised manuscript we have added a paragraph about our opinion concerning the use and future perspective of depth of hypnosis monitoring in children.

Kind regards,

Y.M. Cheung, M.D.